////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-Deque]]++Deque++::
* ++[[painless-api-reference-Deque-addFirst-1]]void link:{java8-javadoc}/java/util/Deque.html#addFirst%2Djava.lang.Object%2D[addFirst](def)++ (link:{java9-javadoc}/java/util/Deque.html#addFirst%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-Deque-addLast-1]]void link:{java8-javadoc}/java/util/Deque.html#addLast%2Djava.lang.Object%2D[addLast](def)++ (link:{java9-javadoc}/java/util/Deque.html#addLast%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-Deque-descendingIterator-0]]<<painless-api-reference-Iterator,Iterator>> link:{java8-javadoc}/java/util/Deque.html#descendingIterator%2D%2D[descendingIterator]()++ (link:{java9-javadoc}/java/util/Deque.html#descendingIterator%2D%2D[java 9])
* ++[[painless-api-reference-Deque-getFirst-0]]def link:{java8-javadoc}/java/util/Deque.html#getFirst%2D%2D[getFirst]()++ (link:{java9-javadoc}/java/util/Deque.html#getFirst%2D%2D[java 9])
* ++[[painless-api-reference-Deque-getLast-0]]def link:{java8-javadoc}/java/util/Deque.html#getLast%2D%2D[getLast]()++ (link:{java9-javadoc}/java/util/Deque.html#getLast%2D%2D[java 9])
* ++[[painless-api-reference-Deque-offerFirst-1]]boolean link:{java8-javadoc}/java/util/Deque.html#offerFirst%2Djava.lang.Object%2D[offerFirst](def)++ (link:{java9-javadoc}/java/util/Deque.html#offerFirst%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-Deque-offerLast-1]]boolean link:{java8-javadoc}/java/util/Deque.html#offerLast%2Djava.lang.Object%2D[offerLast](def)++ (link:{java9-javadoc}/java/util/Deque.html#offerLast%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-Deque-peekFirst-0]]def link:{java8-javadoc}/java/util/Deque.html#peekFirst%2D%2D[peekFirst]()++ (link:{java9-javadoc}/java/util/Deque.html#peekFirst%2D%2D[java 9])
* ++[[painless-api-reference-Deque-peekLast-0]]def link:{java8-javadoc}/java/util/Deque.html#peekLast%2D%2D[peekLast]()++ (link:{java9-javadoc}/java/util/Deque.html#peekLast%2D%2D[java 9])
* ++[[painless-api-reference-Deque-pollFirst-0]]def link:{java8-javadoc}/java/util/Deque.html#pollFirst%2D%2D[pollFirst]()++ (link:{java9-javadoc}/java/util/Deque.html#pollFirst%2D%2D[java 9])
* ++[[painless-api-reference-Deque-pollLast-0]]def link:{java8-javadoc}/java/util/Deque.html#pollLast%2D%2D[pollLast]()++ (link:{java9-javadoc}/java/util/Deque.html#pollLast%2D%2D[java 9])
* ++[[painless-api-reference-Deque-pop-0]]def link:{java8-javadoc}/java/util/Deque.html#pop%2D%2D[pop]()++ (link:{java9-javadoc}/java/util/Deque.html#pop%2D%2D[java 9])
* ++[[painless-api-reference-Deque-push-1]]void link:{java8-javadoc}/java/util/Deque.html#push%2Djava.lang.Object%2D[push](def)++ (link:{java9-javadoc}/java/util/Deque.html#push%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-Deque-remove-1]]boolean link:{java8-javadoc}/java/util/Deque.html#remove%2Djava.lang.Object%2D[remove](def)++ (link:{java9-javadoc}/java/util/Deque.html#remove%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-Deque-removeFirst-0]]def link:{java8-javadoc}/java/util/Deque.html#removeFirst%2D%2D[removeFirst]()++ (link:{java9-javadoc}/java/util/Deque.html#removeFirst%2D%2D[java 9])
* ++[[painless-api-reference-Deque-removeFirstOccurrence-1]]boolean link:{java8-javadoc}/java/util/Deque.html#removeFirstOccurrence%2Djava.lang.Object%2D[removeFirstOccurrence](def)++ (link:{java9-javadoc}/java/util/Deque.html#removeFirstOccurrence%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-Deque-removeLast-0]]def link:{java8-javadoc}/java/util/Deque.html#removeLast%2D%2D[removeLast]()++ (link:{java9-javadoc}/java/util/Deque.html#removeLast%2D%2D[java 9])
* ++[[painless-api-reference-Deque-removeLastOccurrence-1]]boolean link:{java8-javadoc}/java/util/Deque.html#removeLastOccurrence%2Djava.lang.Object%2D[removeLastOccurrence](def)++ (link:{java9-javadoc}/java/util/Deque.html#removeLastOccurrence%2Djava.lang.Object%2D[java 9])
* Inherits methods from ++<<painless-api-reference-Collection,Collection>>++, ++<<painless-api-reference-Iterable,Iterable>>++, ++<<painless-api-reference-Object,Object>>++, ++<<painless-api-reference-Queue,Queue>>++
